Author’s response to reviews

Title: A process evaluation of a stroke-specific follow-up care model for stroke patients and caregivers; a longitudinal study.

Authors:

Manon Fens (m.fens@mumc.nl)
George Beusmans (george.beusmans@maastrichtuniversity.nl)
Martien Limburg (martienlimburg@gmail.com)
Liesbeth van Hoef (liesbeth.van.hoef@mumc.nl)
Jolanda van Haastregt (j.vanhaastregt@maastrichtuniversity.nl)
Job Metsemakers (job.metsemakers@maastrichtuniversity.nl)
Caroline van Heugten (c.vanheugten@maastrichtuniversity.nl)

Version: 5
Date: 15 November 2014

Author’s response to reviews: see over
Dear Editor,

Thank you for the opportunity to re-revise our manuscript, entitled: ‘A process evaluation of a stroke-specific follow-up care model for stroke patients and caregivers; a longitudinal study’ (MS: 2116970981112478). We believe that the comments of the reviewer about the ‘SCCs and nurses’ and revising our manuscript accordingly have improved our paper. In response to your email, we have revised our manuscript according to your requests. The changes of the re-revision are marked pink in the manuscript, in addition to the yellow markings of the previous revision.

- Based on the question of the reviewer, we have explained in the manuscript that the SCCs were indeed nurses, who were specialised in stroke and in long-term care after stroke (page 6, lines 9-10)
- The nurses were indeed the SCCs who registered the follow-up care. We have changed the ‘nurses’ into ‘SCCs’ in the manuscript (page 7, line 25).

We hope these re-revisions are to your satisfaction and that you can consider our manuscript for publication in your journal now. We look forward to your decision.

Thank you and kind regards,

On behalf of all authors,

Caroline van Heugten

Maastricht University
Department of Psychiatry and Neuropsychology
PO Box 616
6200 MD Maastricht
The Netherlands
T 0031 43 3884091
F 0031 43 3884092
c.vanheugten@maastrichtuniversity.nl